Effects of Empagliflozin on Fluid Overload, Weight, and Blood Pressure in CKD

Kaitlin J. Mayne (),^{1,2} Natalie Staplin (),¹ David F. Keane (),³ Christoph Wanner (),¹

Susanne Brenner (),⁴ Vladimir Cejka (),⁴ Johannes Stegbauer (),^{5,6}

Parminder K. Judge (b, ^{1,7} David Preiss (b, ¹ Jonathan Emberson (b, ¹ Daniele Trinca (b, ⁸ Rejive Dayanandan,¹ Ryonfa Lee (b, ¹ John Nolan,¹ Akiko Omata,¹ Jennifer B. Green (b, ⁹ David Z. I. Cherney,¹⁰ Lai Seong Hooi (b, ¹¹ Roberto Pontremoli (b, ¹² Katherine R. Tuttle (b, ¹³ Jennifer S. Lees (b, ² Patrick B. Mark (b, ² Simon J. Davies (b, ¹⁴ Sibylle J. Hauske (b, ^{15,16} Dominik Steubl, ^{15,17,18} Martina Brückmann (b, ^{15,19} Martin J. Landray,¹ Colin Baigent,¹ Richard Haynes (b, ^{1,7} and

William G. Herrington,^{1,7}on behalf of the EMPA-KIDNEY Collaborative Group*

Due to the number of contributing authors, the affiliations are listed at the end of this article.

ABSTRACT

Background CKD is associated with fluid excess that can be estimated by bioimpedance spectroscopy. We aimed to assess effects of sodium glucose co-transporter 2 inhibition on bioimpedance-derived "Fluid Overload" and adiposity in a CKD population.

Methods EMPA-KIDNEY was a double-blind placebo-controlled trial of empagliflozin 10 mg once daily in patients with CKD at risk of progression. In a substudy, bioimpedance measurements were added to the main trial procedures at randomization and at 2- and 18-month follow-up visits. The substudy's primary outcome was the study-average difference in absolute "Fluid Overload" (an estimate of excess extracellular water) analyzed using a mixed model repeated measures approach.

Results The 660 substudy participants were broadly representative of the 6609-participant trial population. Substudy mean baseline absolute "Fluid Overload" was 0.4 ± 1.7 L. Compared with placebo, the overall mean absolute "Fluid Overload" difference among those allocated empagliflozin was -0.24 L (95% confidence interval [CI], -0.38 to -0.11), with similar sized differences at 2 and 18 months, and in prespecified subgroups. Total body water differences comprised between-group differences in extracellular water of -0.49 L (95% CI, -0.69 to -0.30, including the -0.24 L "Fluid Overload" difference) and a -0.30 L (95% CI, -0.57 to -0.03) difference in intracellular water. There was no significant effect of empagliflozin on bioimpedance-derived adipose tissue mass (-0.28 kg [95% CI, -1.41 to 0.85]). The between-group difference in weight was -0.7 kg (95% CI, -1.3 to -0.1).

Conclusions In a broad range of patients with CKD, empagliflozin resulted in a sustained reduction in a bioimpedance-derived estimate of fluid overload, with no statistically significant effect on fat mass.

Trial Registration Clinicaltrials.gov: NCT03594110; EuDRACT: 2017-002971-24 (https://eudract.ema.europa.eu/).

JASN 35: 202-215, 2024. doi: https://doi.org/10.1681/ASN.00000000000271

This is an open access article distributed under the Creative Commons Attribution License 4.0 (CCBY), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

INTRODUCTION

Patients with CKD are at increased risk of cardiovascular disease,^{1,2} key features of which are structural heart disease, heart failure, and sudden death.³⁻⁵ These risks increase progressively as eGFR decreases,6 with risk of death from cardiovascular disease exceeding risk of progression to kidney failure for many people with CKD. Fluid excess is common in CKD, especially when heart failure coexists,7 and can be quantified using bioimpedance spectroscopy.⁸ Bioimpedance can estimate a number of fluid- and adiposity-related parameters, including the excess constituent of total body extracellular water (ECW) over and above what is considered normohydration. We refer to this parameter as "Fluid Overload" (refer to Figure 1 and the Supplemental Methods for more details about bioimpedance spectroscopy and a glossary of fluid-related terms).9 "Fluid Overload" can be used to guide dialysis prescription,¹⁰ and epidemiologically there are positive associations between bioimpedance-measured "Fluid Overload" with cardiovascular outcomes and mortality in patients on dialysis, with nondialysis CKD, or with heart failure.8

The double-blind international multicenter Study of Heart and Kidney Protection With Empagliflozin (EMPA-KIDNEY) demonstrated that, compared with matching placebo, empagliflozin 10 mg once daily reduced the risk of kidney disease progression or cardiovascular death by 28% (95% confidence interval [CI], 18% to 36%) in 6609 patients with CKD at risk of progression.¹¹ A meta-analysis of large placebo-controlled trials extended these findings and showed that in people with CKD, heart failure, or type 2 diabetes at high cardiovascular risk, sodium glucose co-transporter 2 (SGLT2) inhibitors safely reduce the risk of kidney disease progression by about two fifths and AKI by about a quarter, with consistent effects irrespective of diabetes status.¹² SGLT2 inhibitors also reduce the risk of cardiovascular outcomes, particularly hospitalization for heart failure.¹² These cardiovascular benefits are particularly large in patients with preexisting heart failure,^{12,13} but smaller numbers of cardiovascular events in patients with CKD without diabetes and at low levels of eGFR mean effects are less certain in these populations.^{11,12} The amount of glycosuria induced by SGLT2 inhibition falls with decreasing eGFR and with ambient normoglycemia,14 so it is reasonable to hypothesize that

Received: August 9, 2023 Accepted: October 26, 2023. Published Online Ahead of Print: December 12, 2023.

*See Supplemental Materials for full list of the coordinating centers and site staff from the EMPA-KIDNEY Collaborative Group who contributed to the substudy.

Copyright $\textcircled{\sc 0}$ 2023 The Author(s). Published by Wolters Kluwer Health, Inc. on behalf of the American Society of Nephrology.

Significance Statement

SGLT2 inhibitors reduce risk of kidney progression, AKI, and cardiovascular disease, but the mechanisms of benefit are incompletely understood. Bioimpedance spectroscopy can estimate body water and fat mass. One quarter of the EMPA-KIDNEY bioimpedance substudy CKD population had clinically significant levels of bioimpedance-derived "Fluid Overload" at recruitment. Empagliflozin induced a prompt and sustained reduction in "Fluid Overload," irrespective of sex, diabetes, and baseline N-terminal pro B-type natriuretic peptide or eGFR. No significant effect on bioimpedance-derived fat mass was observed. The effects of SGLT2 inhibitors on body water may be one of the contributing mechanisms by which they mediate effects on cardiovascular risk.

other effects of SGLT2 inhibitors could also be attenuated in such patients.^{11,15} To address uncertainty about the effects of SGLT2 inhibitors on fluid status and adiposity in CKD, we embedded a bioimpedance-based substudy within the EMPA-KIDNEY trial.¹¹ The primary aim was to assess the effects of empagliflozin 10 mg once daily versus placebo on fluid status using the bioimpedance-derived parameter of absolute "Fluid Overload" (*i.e.*, estimated excess ECW). We also aimed to assess effects on this "Fluid Overload" parameter over time and in different types of patients with CKD. In this report, we also put the substudy findings regarding empagliflozin's effects on bioimpedance-derived fluid and adiposity parameters in the context of its potentially related effects on weight, BP, glycated hemoglobin and hematocrit (as observed in the full trial cohort).

METHODS

Substudy Design and Population

The full methods of the EMPA-KIDNEY trial and the main results have been reported elsewhere (ClinicalTrials.gov number, NCT03594110; EudraCT number, 2017-002971-24).11,16 In brief, patients with CKD at risk of progression were identified based on historical and screening local laboratory measurements of an eGFR \geq 20 but <45 ml/min per 1.73 m², or an eGFR \geq 45 but <90 ml/min per 1.73 m² with a urinary albumin-creatinine ratio (uACR) \geq 200 mg/g. This report details the results of an optional substudy conducted in a subset of sites in the United Kingdom and Germany which added bioimpedance measurements at the randomization, 2- and 18-month follow-up visits to the trial's main protocolspecified procedures (substudy protocol supplement available in the Supplemental Materials). All participants provided written informed consent. Regulatory authorities, as well as ethics committees at each center, approved the trial and the substudy which adhere to the Declaration of Helsinki.

Bioimpedance Measurements

Bioimpedance spectroscopy is a tool used in the clinical care of patients requiring dialysis to monitor fluid status.¹⁷ We

Correspondence: William G. Herrington, Medical Research Council Population Health Research Unit at the University of Oxford, Clinical Trial Service Unit and Epidemiological Studies Unit (CTSU), Nuffield Department of Population Health, University of Oxford, Richard Doll Building, Old Road Campus, Roosevelt Drive, Oxford OX3 7LF, United Kingdom. Email: will.herrington@ndph.ox.ac.uk.



Figure 1. Relationship of the derived "Fluid Overload" parameter to body weight and tissue mass. Based on the threecompartment model described by Chamney *et al.*⁹ *Excess ECW accumulates both in tissues and in the blood (although blood volume is not specifically conceptualized in the three-compartment model), so changes in Fluid Overload could reflect changes in excess ECW that might be residing in adipose tissue, lean tissue, or both. [†]Refers to normally hydrated lean and adipose tissue mass. Total body water (TBW) is the sum of ECW and ICW, although TBW is not conceptualized in the three-compartment model. The figure is not to scale because compartment proportions vary between individuals and "Fluid Overload" is usually smaller than depicted (and can be a negative value in fluid depletion). The mean baseline values in the EMPA-KIDNEY substudy were total body weight 88.8 kg, "Fluid Overload" 0.4 L, lean tissue mass 38.8 kg, and adipose tissue mass 49.6 kg. In the EMPA-KIDNEY substudy, mean total ECW at baseline was 18.7 L and ICW 20.4 L. ECW, extracellular water; ICW, intracellular water.

used the Fresenius Medical Care Body Composition Monitor (BCM) bioimpedance spectroscopy device because it has been extensively validated for fluid status assessment in kidney failure populations and used in randomized controlled trials.^{18–20} The device passes low-level electrical current at frequencies of 5–1000 kHz (with results extrapolated from zero to infinity kHz) between electrodes attached to patients' hands and feet.⁸ All substudy bioimpedance measurements were performed by trained local research coordinators. Body fluid and adiposity indices were then derived centrally using age, sex, a paired weight measurement, and height data combined with bioimpedance measurements of electrical resistance, and a validated three-compartment model formula using proprietary coefficients.^{9,21}

The primary outcome was based on the bioimpedancederived estimate of excess ECW which we refer to as absolute "Fluid Overload" (sometimes referred to as "overhydration"). It is reported in liters and can have positive or negative values (Figure 1). Its reference range estimated from the 10th and 90th centiles of a reference general population distribution is -1.1 L to +1.1 L.²² "Fluid Overload" can be indexed to ECW volume and referred to as percentage relative "Fluid Overload." An absolute value of +1.1 L approximately corresponds to relative "Fluid Overload" of +7%.²³ Values above this threshold have been consistently associated with an increased risk of death and cardiovascular events,⁸ and we refer to it as moderate "Fluid Overload" (>7%, \leq 15%) or severe "Fluid Overload" (>15%).^{8,23,24} Bioimpedance measurements were also used to derive estimates of extracellular and intracellular water (ICW) volume, lean tissue index (LTI), and fat tissue index (FTI) (see Supplemental Methods for more details).

Local research coordinators were trained to repeat measurements when the BCM device's automated quality score (the Q value) was below 80 (out of 100). Visual inspection of reactance versus resistance plots (known as Cole–Cole plots) were additionally used to assess data quality.²⁵ It was not always possible to obtain a Q value \geq 80, so any measurement with a Q value <80 had its Cole–Cole plot assessed independently by two researchers to determine data quality and inclusion in the primary assessment using prespecified rules blind to treatment allocation (see prespecified Data Analysis Plan provided in the Supplemental Materials for details). Absolute "Fluid Overload" values lower than -5 L were consistently associated with low-quality bioimpedance measurement and were considered invalid.

Outcomes

The substudy's prespecified primary outcome was the effect of empagliflozin versus placebo on mean absolute "Fluid Overload" averaged over time, with effects on relative "Fluid Overload" provided for completeness. It was estimated that at least 382 participants would provide >90% power (at a twosided P value of 0.05) to detect at least a 0.3-L difference in absolute "Fluid Overload" between treatment groups. The key secondary outcome was the effect of empagliflozin versus placebo on time to the first event of a cardiovascular composite defined as death from heart failure, heart failure hospitalization, or development of new moderate or severe "Fluid Overload" (in participants without this level of "Fluid Overload" at baseline). The other secondary outcomes were the effects of empagliflozin versus placebo on "Fluid Overload" at the different measurement time points. Tertiary assessments are detailed in the Supplemental Methods and include analyses of the effects of empagliflozin versus placebo on all ECW (of which "Fluid Overload" is a constituent) and ICW. In addition, the effects of empagliflozin versus placebo on total body water (the sum of all ECW and ICW) were assessed as a post hoc analysis to contextualize effects on "Fluid Overload."

In order for inferences from the bioimpedance substudy to be put in the context of findings from all the available EMPA-KIDNEY data, additional analyses included assessments of the effects of empagliflozin versus placebo on weight, body mass index (BMI), waist-to-hip ratio, glycated hemoglobin, hematocrit, and BP (systolic and diastolic) in the full trial cohort. Analyses emphasized results of studyaverage effects including all available measurements from routine trial visit time points (with effects at 2 and 18 months also presented). The full cohort results are emphasized because of greater statistical power and wider generalizability than the substudy. Substudy results were compared with results from the full cohort using standard statistical tests of heterogeneity. Analyses of weight and systolic BP also considered results for the same subgroups as the substudy (plus self-reported race, to explore effects by race in the full trial cohort because the substudy took place in the United Kingdom and Germany only). Prespecified sensitivity analysis for the primary outcome included three analyses assessing any effect of data quality assessments. Analyses of effects of empagliflozin on diuretic use were included post hoc.

Statistical Analysis

Substudy analyses followed the intention-to-treat principle and required a consenting participant to have provided at least one valid bioimpedance measurement. The primary outcome was prespecified to be assessed using a mixed model repeated measures (MMRM) approach adjusted for age, sex, previous diabetes, eGFR, and uACR in the categories used in the minimized randomization algorithm.¹¹ The MMRM model also included fixed categorical effects of time (to avoid assuming a linear association between treatment allocation and "Fluid Overload" over time), treatment allocation, and treatment-by-time interaction, and continuous effects of baseline (randomization) measurements and baseline-by-time interaction. The within-person error correlations were assumed to be unstructured. Analyses of the full trial cohort were additionally adjusted for region.¹¹ Effects at each follow-up time point were estimated and used to derive study-average effects (with weights proportional to the amount of time between visits). All betweengroup differences are reported as empagliflozin minus placebo. To assess effect modification, subgroup-specific treatment effects were estimated by fitting interaction terms in the MMRM models. The null hypothesis was that the treatment effect is the same across all subgroups. This was tested by calculating a heterogeneity or trend statistic from subgroupspecific means and standard errors, without correction for multiplicity of testing.

The key secondary outcome and its components were analyzed using an adjusted Cox proportional hazards regression using the same covariates in the minimization algorithm (age, sex, previous diabetes, eGFR, and uACR) and included the complete substudy population of 660 participants (i.e., it included participants without a valid follow-up bioimpedance measurement who were excluded from MMRM analyses but were at risk of clinical outcomes). Tertiary analyses used the same MMRM approach as described for the primary outcome and assessed effects on ECW, ICW, LTI, FTI, body weight, and BMI. Waist and hip circumference measurements were obtained at a single follow-up time point (18 months) and were therefore analyzed by analysis of covariance, adjusted for the baseline value and minimization variables. Handling of missing data is outlined in the Supplemental Methods. P values for hypothesis testing for outcomes are limited to the primary outcome. P values for testing for any evidence of effect modification between subgroups, and between treatment effect and effects by time are provided. The prespecified Data Analysis Plan is provided in the Supplemental Materials. Analyses were performed using R Studio version 4.2.2 (RStudio: Integrated Development for R. RStudio, PBC, Boston, MA) and SAS version 9.4 (SAS Institute, Cary, NC).

RESULTS

Substudy Baseline Characteristics and Adherence

Between May 22, 2019, and April 14, 2021, 668 participants consented to join the substudy. One was excluded because of a metal knee implant and no useable bioimpedance measurement at baseline excluded a further seven, leaving 660 included in analyses (Supplemental Figure 1, Supplemental Material). MMRM analyses excluded 40 consenting participants with no valid follow-up bioimpedance measurement (empagliflozin versus placebo: 21 versus 19, respectively; three due to death before first follow-up measurement, 28 with no follow-up measurement performed [*e.g.*, due to coronavirus disease 2019 precluding visits], and nine due to low data

Table 1.	Bioim	pedance	substudy	cohort:	baseline	characteristics
----------	-------	---------	----------	---------	----------	-----------------

Baseline Characteristic	Empagliflozin (n=332)	Placebo (n=328)
Demographics		
Age (yr)	65.2 (14.2)	64.1 (14.9)
Female sex	102 (30.7)	103 (31.4)
White race	321 (96.7)	315 (96.0)
Previous disease		
Diabetes	135 (40.7)	121 (36.9)
Heart failure	62 (18.7)	74 (22.6)
Clinical measurements		
Weight (kg)	89.8 (20.2)	87.9 (19.3)
Body mass index (kg/m ²)	30.5 (6.2)	30.1 (6.3)
Waist-to-hip ratio	1.0 (0.1)	1.0 (0.1)
Systolic BP (mm Hg)	137.0 (18.8)	137.5 (18.9)
Diastolic BP (mm Hg)	77.8 (12.2)	78.6 (11.9)
Bioimpedance measurements ^a		
Absolute "Fluid Overload" (L)	0.45 (1.68)	0.32 (1.68)
Relative "Fluid Overload" (%)		
Mean (SD)	1.9 (8.7)	1.3 (8.3)
Moderate "Fluid Overload"	70 (21.1)	56 (17.1)
Severe "Fluid Overload"	14 (4.2)	16 (4.9)
Extracellular water (L)	19.0 (3.8)	18.4 (3.7)
Intracellular water (L)	20.7 (4.5)	20.1 (4.6)
Lean tissue index (kg/m²)	13.3 (3.1)	12.9 (3.0)
Fat tissue index (kg/m²)	12.6 (5.4)	12.5 (5.1)
Laboratory measurements		
eGFR (ml/min per 1.73 m²)		
Mean (SD)	36.1 (13.4)	35.8 (11.4)
Distribution		
<30	123 (37.0)	118 (36.0)
≥30 <45	148 (44.6)	154 (47.0)
≥45	61 (18.4)	56 (17.1)
Urinary albumin-creatinine ratio (mg/g)	203 (26–958)	205 (29–865)
HbA1c (mmol/mol)	43.9 (11.3)	43.5 (10.9)
NT-proBNP (ng/L)	197 (90–596)	225 (95–550)
Medications		
RAS inhibitor	304 (91.6)	288 (87.8)
Any diuretic therapy	180 (54.2)	173 (52.7)

Data are presented as mean (SD) or median (Q1–Q3) for continuous variables and *n* (%) for categorical variables. HbA1c, glycated hemoglobin; NT-proBNP, N-terminal pro–brain-type natriuretic peptide; RAS, renin-angiotensin system.

^aBioimpedance measurements are presented for 644 of 660 participants with a baseline measurement (missing for 16/660) irrespective of validity for inclusion in the primary analysis.

quality). This left a total of 620 participants for whom 1047 valid follow-up bioimpedance measurements were available for MMRM analyses.

In the substudy, the mean age was 64 (15) years and 205 participants (31%) were female (Table 1). At recruitment, 136 (21%) reported a diagnosis of heart failure and 256 (39%) had diabetes. The mean (SD) eGFR was 36.0 (12.4) ml/min per 1.73 m² and median (Q1–Q3) N-terminal pro–B-type natriuretic peptide (NT-proBNP) was 211 (93–581) ng/L. The mean body weight was 88.8 (19.8) kg and mean BMI was 30.3 (6.2) kg/m². The mean absolute "Fluid Overload" at baseline was 0.4 (1.7) L with 126 (19%) and 30 (5%) participants with evidence of moderate and severe "Fluid Overload," respectively (Table 1). Severity of "Fluid Overload" mirrored established markers of fluid excess: heart failure was twice as common in those with severe "Fluid Overload" compared with the normohydrated group and NT-proBNP was five-fold higher (Supplemental Table 2). In addition, participants with

"Fluid Overload" were more likely to be older, be male, to have previous diabetes, and have a lower eGFR (Supplemental Table 2). The substudy cohort characteristics were broadly representative of the full trial cohort,¹¹ although were less racially diverse due to being conducted only in the United Kingdom and Germany (Supplemental Table 3).

Substudy adherence to study treatment was consistent with adherence in the full-trial population.¹¹ At 12 months of follow-up (the approximate midpoint of the trial), of substudy participants who remained alive, 282 of 318 (88.7%) in the empagliflozin group and 292 of 320 (91.3%) in the placebo group reported taking at least 80% of their allocated study treatment.

Effects on Bioimpedance-Derived Parameters

The primary assessment found that the study-average mean absolute "Fluid Overload" was 0.24 L lower in those allocated to the empagliflozin group compared with the placebo group



Figure 2. Effects of empagliflozin on mean bioimpedance-derived absolute "Fluid Overload" by time. The value at time 0 is the average across all randomized participants. Follow-up means (and their Cls) are derived from a repeated measures mixed model adjusted for baseline values, age, sex, diabetes, eGFR, and uACR. Follow-up values are plotted at the median follow-up day in each time window. There was no significant interaction between treatment allocation and time (P = 0.11). The study average is the between-group difference (empagliflozin minus placebo) in weighted averages of both time points (see Supplemental Methods). Analyses excluded 40 consenting participants with no valid follow-up measurements. Median (Q1–Q3) follow-up since randomization for empagliflozin versus placebo groups at the 2-month visit: 64 (57–74) versus 64 (57–75) days, Wilcoxon rank sum P = 0.871; and at the 18-month visit: 540 (519–555) versus 532 (505–554) days, P = 0.026. Cl, confidence interval; uACR, urinary albumin-creatinine ratio.

(absolute difference in means -0.24 L, 95% CI, -0.38 to -0.11), with similar differences at 2 months (-0.23 L, 95% CI, -0.37 to -0.08) and 18 months (-0.26 L, 95% CI, -0.46 to -0.06) (Figure 2, Table 2). Findings were robust in sensitivity analyses assessing the effect of data quality assessments (Supplemental Table 4). The effect of empagliflozin on the primary outcome was similar in subgroups by sex, diabetes status, and across the spectrum of NT-proBNP and eGFR studied (*P* values for heterogeneity or trend >0.3, Figure 3 and Supplemental Table 5). Neither was there any evidence of heterogeneity in *post hoc* exploratory subgroups divided by baseline fluid status (fluid depletion, low and high normohydration, and moderate and severe "Fluid Overload"; P = 0.71), diuretic use (P = 0.07), or uACR (P = 0.33, Supplemental Figure 2).

There was no significant difference in the composite outcome between treatment groups (empagliflozin 35/332 [11%] versus placebo 38/328 [12%], hazard ratio (HR) 0.91, 95% CI, 0.57 to 1.45, P = 0.69) with consistent effects for its components (Table 3). The number of outcomes was low, limiting statistical power: development of new moderate "Fluid Overload" occurred in 7.8% of substudy participants allocated empagliflozin versus 10.1% allocated placebo, and development of new severe "Fluid Overload" occurred in 2.6% versus 1.3% of empagliflozin and placebo groups, respectively. The tertiary outcome of regression of moderate or severe "Fluid Overload" did not differ significantly between the empagliflozin and placebo groups (54.8% versus 48.6%; Table 3). Heart failure events were also infrequent; there were no deaths due to heart failure in

the substudy population. In the full trial cohort, hospitalization for heart failure occurred in 2.7% and 3.2% of participants allocated empagliflozin and placebo, respectively (HR 0.80, 95% CI, 0.60 to 1.06), and findings from the substudy cohort considered in isolation were consistent (empagliflozin 3.3% versus placebo 4.9%; HR 0.67, 95% CI, 0.31 to 1.46; Table 3).

Bioimpedance estimated that the study-average absolute difference in total body water was -0.82 L (-1.24 to -0.40). This consisted of differences in ECW of -0.49 L (95% CI, -0.69 to -0.30) (of which the -0.24 L between-group difference in "Fluid Overload" is a constituent) and ICW of -0.30 L (95% CI, -0.57 to -0.03). There were no significant between-group differences in bioimpedance-derived fat or lean tissue index or related tissue mass parameters (lean, fat, and adipose tissue mass in kg; Table 2, Supplemental Tables 6 and 7). In the bioimpedance substudy population, the study-average between-group difference in weight was -0.7 kg (-1.3 to -0.1). Supplemental Figure 3 shows the change in weight (relative to baseline) with the change in different biompedance indices at the 2-month follow-up visit.

Effects on Anthropometry, BP, and Relevant Laboratory Values in the Full Trial Cohort

In the full trial cohort, the between-group difference in weight was -0.9 kg (95% CI, -1.2 to -0.6) (Figure 4, Supplemental Table 8) and the effect of empagliflozin on weight did not vary significantly over time (interaction *P* value by time = 0.47, Supplemental Table 8). In the full cohort, there was no

Table 2.	Effects of	empagliflozin	on bioim	pedance-derived	parameters

	Empagliflozir	n (<i>n</i> =311)	Placebo (n	=309)	A		P Value for
Bioimpedance-Derived Parameter	Adjusted ^a Mean	idjusted ^a SE Adjusted ^a SE Difference Mean SE Mean		Absolute Difference	95% CI	Primary Outcome	
Primary assessments							
Absolute "Fluid Overload", L							
Study average	0.10	0.05	0.34	0.05	-0.24	(-0.38 to -0.11)	< 0.001
Relative "Fluid Overload", %							
Study average	0.14	0.25	1.33	0.25	-1.19	(−1.90, to −0.48)	0.001
Secondary assessments							
Absolute "Fluid Overload", L							
Randomization	0.50	0.09	0.35	0.09			
2-mo follow-up	0.18	0.05	0.40	0.05	-0.23	(-0.37 to -0.08)	
18-mo follow-up	0.01	0.07	0.27	0.07	-0.26	(-0.46 to -0.06)	
Relative "Fluid Overload", %							
Randomization	2.24	0.47	1.39	0.45			
2-mo follow-up	0.52	0.27	1.65	0.27	-1.12	(−1.88 to −0.37)	
18-mo follow-up	-0.36	0.38	0.92	0.37	-1.28	(-2.32 to -0.23)	
Tertiary assessments						· · ·	
Extracellular water, L							
Study average	18.16	0.07	18.66	0.07	-0.49	(-0.69 to -0.30)	
Intracellular water, L						· · · ·	
Study average	20.10	0.10	20.40	0.10	-0.30	(-0.57 to -0.03)	
Lean tissue index (LTI), kg/m ²							
Study average	12.90	0.09	13.05	0.09	-0.14	(-0.39 to 0.10)	
Fat tissue index (FTI), kq/m^2							
Study average	12.34	0.10	12.42	0.10	-0.07	(-0.35 to 0.20)	

CI, confidence interval; SE, standard error.

^aMean effects are adjusted for baseline values of the dependent variable (in continuous form) and for any differences in key baseline characteristics (categories of age, sex, diabetes, eGFR, and urinary albumin-creatinine ratio) between treatment groups with study averages weighted in proportion to the amount of time between follow-up visits (see Supplemental Methods). Analysis excluded 40 consenting participants with no valid follow-up measurements (3 deaths before first follow-up measurement, 28 with no measurement performed, and 9 excluded because of inadequate data quality). Effects on "Fluid Overload" did not vary by time: *P* value for interaction with time=0.11 and 0.39 for absolute and relative "Fluid Overload", respectively.

evidence of heterogeneity of the effect of empagliflozin on weight in subgroups by sex, baseline eGFR, or diabetes (Figure 4, or in *post hoc* analyses by race: Supplemental Figure 4). The waist-to-hip ratio at 18 months was also not significantly different between the empagliflozin versus placebo groups (Supplemental Table 9). The study-average difference in glycated hemoglobin (HbA1c) in the full cohort was -0.4 mmol/mol (95% CI, -0.8 to -0.0), with a -0.9 mmol/mol (95% CI, -1.6 to -0.1) difference in HbA1c in participants with diabetes at randomization and no significant difference in participants without diabetes (0.0 mmol/mol, 95% CI, -0.2 to 0.2; Supplemental Table 10). The full trial cohort average between-group difference in hematocrit at 18 months postrandomization was 2.3% (95% CI, 1.9 to 2.7).

The study-average between-group differences in systolic and diastolic BP were -2.6 (95% CI, -3.3 to -1.9) and -0.5 mm Hg (95% CI, -0.9 to -0.1), respectively. In the full trial cohort, there was no evidence of heterogeneity of the effect of empagliflozin on systolic BP when subdivided by sex, baseline eGFR, NT-proBNP (Figure 4), or race (Supplemental Figure 4), but there was some evidence to suggest a larger systolic BP difference in patients with diabetes (Figure 4). Effects on anthropometry, HbA1c, hematocrit, and BP in the substudy were approximately consistent with the full trial cohort results (Supplemental Tables 8–11).

Effects on Diuretic Use

Among those participants in the full trial cohort who were not taking a loop diuretic at randomization, 159 of 2453 (6.5%) in the empagliflozin group compared with 212 of 2409 (8.8%) in the placebo group started such medication during follow-up, representing a 26% lower likelihood of a new loop diuretic prescription among the empagliflozin group (risk ratio 0.74, 95% CI, 0.60 to 0.90).

DISCUSSION

In the EMPA-KIDNEY substudy of 660 patients with CKD, empagliflozin resulted in a sustained reduction in bioimpedancederived "Fluid Overload" for at least 18 months, irrespective of diabetes status or level of kidney function. Using the threecompartment model, we observed a -0.24 L between-group difference in "Fluid Overload" but no significant differences in normally hydrated lean or adipose tissue compartments. Fluid volume differences consisted of approximately 0.8 L less total body water of which approximately 0.5 L was ECW and approximately 0.3 L ICW (with the approximately 0.5 L total ECW difference including the -0.24 L between-group difference in excess ECW referred to as "Fluid Overload"). These data raise a hypothesis that an important determinant of the

Subgroup	Baseline Mean (SE)	Difference (95% CI)	P _{het/trend}
Sex			0.93
Male	0.64 (0.08)	-0.24 (-0.40, -0	.08)
Female	-0.05 (0.10)	-0.25 (-0.50, -0	.00)
Diabetes			0.38
Absent	0.18 (0.07)	-0.19 (-0.36, -0	.02)
Present	0.83 (0.11)	-0.32 (-0.54, -0	.10)
NT-proBNP, ng	g/L		0.82
<110	-0.33 (0.10)	-0.36 (-0.61, -0	.10)
≥110 <330	0.22 (0.09)	-0.07 (-0.30, 0	.15)
≥330	1.30 (0.11)	-0.30 (-0.53, -0	.07)
Estimated GF	R, ml/min/1.73m ²		0.33
<30	0.72 (0.11)	-0.11 (-0.34, 0	.12)
≥30 <45	0.22 (0.09)	-0.30 (-0.50, -0	.11)
≥45	0.36 (0.15)	-0.27 (-0.59, 0	.05)
Overall	0.43 (0.06)	-0.24 (-0.38, -0	.11)
	-1.0	-0.5 0 0.5	
	,	、 、	

Favors Empagliflozin Favors Placebo

Figure 3. Effects of empagliflozin on mean bioimpedance-derived absolute "Fluid Overload" (in liters) by prespecified substudy subgroups. Study-average differences are adjusted for baseline values of the dependent variable (in continuous form) and for any differences in key baseline characteristics (categories of age, sex, diabetes, eGFR, and urinary albumin-creatinine ratio) between treatment groups and weighted in proportion to the amount of time between follow-up visits (see Supplemental Methods). Analysis excluded 40 consenting participants with no valid follow-up measurements (3 deaths before first follow-up measurement, 28 with no measurement performed, and 9 excluded because of inadequate data quality). Further details are available in Supplemental Table 5. NT-proBNP, N-terminal pro B-type natriuretic peptide; SE, standard error.

substudy -0.7 kg weight difference was due to effects on fluid status. Along with other mechanisms,²⁶ this effect may contribute to the cardiovascular benefits of SGLT2 inhibitors.

diovascular benefits of SGLT2 inhibitors, but their effects on fluid status in CKD—where effects may be hypothesized to be attenuated by decreased kidney function—have not previously been quantified in randomized trials.^{15,26–28} In patients with type 2 diabetes without kidney disease,

Osmotic diuretic and natriuretic actions are considered potentially important contributing mechanisms to the car-

Table 3.	Effects of empage	iflozin on	cardiovascular	composite	outcome (bioimpedance	substudv	cohort)
10010 01	Encers of empagi		cal allo rabcalai	composito	ourconne (bioinipedance	Substaay	00110110

			-				
	Empagliflozin		Placebo				P Value
Outcome	No. of % No. of % Participants / Total		%	Hazard Ratio	95% CI		
Key secondary assessment							
Death from heart failure, hospitalization for heart	35/332	10.5	38/328	11.6	0.91	(0.57 to 1.45)	0.69
failure, development of new moderate or severe							
"Fluid Overload"							
Death from heart failure	0/332	0.0	0/328	0.0		—	
Hospitalization for heart failure	11/332	3.3	16/328	4.9	0.67	(0.31 to 1.46)	
Development of new moderate "Fluid Overload" ^a	18/232	7.8	25/247	10.1	0.68	(0.37 to 1.26)	
Development of new severe "Fluid Overload" ^b	8/302	2.6	4/303	1.3	1.96	(0.57 to 6.71)	
Tertiary assessment							
Regression of "Fluid Overload" ^c	46/84	54.8	35/72	48.6	1.33	(0.82 to 2.18)	
All and have a time to first suggesting the Company of the	بالمحيد والمام ومراجا والمرجوع المرج	بالألم مراجرات	ومعاميه والمعام والمعام والمعار		· · ·	والمترور والمراجع فالمرجع والمراجع	

All analyses use a time-to-first-event approach. Cox proportional hazards models include adjustment for the covariates used in the minimization algorithm: age, sex, diabetes status, eGFR, and urinary albumin-creatinine ratio. Results were consistent in *post hoc* sensitivity analyses additionally adjusted for use of any diuretic or loop diuretics at baseline (hazard ratios [95% CIs] 0.89 [0.56 to 1.42] and 0.92 [0.58 to 1.47]; respectively). CI, confidence interval.

^aRequires randomization value of relative "Fluid Overload" \leq 7% and follow-up value >7%, \leq 15%.

^bRequires randomization value of relative "Fluid Overload" ≤15% and follow-up value >15%.

^CRequires randomization value consistent with moderate or severe relative "Fluid Overload" and regression to any lower hydration category at any follow-up (limited to first event). All 660 participants were included in the composite outcome analysis because all participants were at risk of the clinical components of the composite. In the full-trial cohort, there were 88 (2.7%) first hospitalizations for heart failure in the empagliflozin group versus 107 (3.2%) in the placebo group: hazard ratio 0.80, 95% CI, 0.60 to 1.06.

Subgroup	Baseline Mean (S	E)	Difference (95% CI)	P _{het/trend}
		WEIGHT (kg)		
Sex				0.56
Male	87.4 (0.3)		-0.8 (-1.2, -0.5)	
Female	76.9 (0.5)	B ¹	-1.0 (-1.5, -0.5)	
Diabetes				0.25
Absent	79.5 (0.3)		-0.8 (-1.1, -0.4)	
Present	89.2 (0.4)		-1.1 (-1.5, -0.7)	
NT-proBNP, ng/L				0.03
<110	82.4 (0.4)		-1.1 (-1.6, -0.6)	
≥110 <330	83.7 (0.5)		-1.2 (-1.7, -0.7)	
≥330	86.2 (0.5)	 	-0.3 (-0.8, 0.2)	
Estimated GFR, ml/min/1.73	m ²			0.26
<30	84.8 (0.4)		-0.8 (-1.3, -0.4)	
≥30 <45	85.0 (0.4)		-0.7 (-1.1, -0.3)	
≥45	80.5 (0.6)		-1.4 (-2.0, -0.8)	
Overall	84.0 (0.3)	\diamond	-0.9 (-1.2, -0.6)	
	s	SYSTOLIC BP (mm Hg)		
Sex		1		0 79
Male	137 4 (0 3)		-26(-35-18)	0.70
Female	134.8 (0.4)		-25 (-36 -1 3)	
Diabetes	104.0 (0.4)		2.0 (0.0, 1.0)	0.001
Absent	134.3 (0.3)		-1.5 (-2.5, -0.6)	0.001
Present	139.2 (0.3)		-3.8 (-4.7, -2.8)	
NT-proBNP, ng/L				0.64
<110	133.2 (0.3)	<u> </u>	-2.1 (-3.31.0)	
≥110 <330	137.6 (0.4)		-3.2 (-4.4, -2.0)	
≥330	139.6 (0.5)		-2.5 (-3.71.2)	
Estimated GFR, mL/min/1.73	3m ²			0.30
<30	137.6 (0.4)	—— —	-3.0 (-4.1, -1.8)	
≥30 <45	136.0 (0.3)		-2.6 (-3.5, -1.6)	
≥45	136.0 (0.5)		-2.0 (-3.5, -0.5)	
Overall	136.5 (0.2)		-2.6 (-3.3, -1.9)	
	. ,	.5 .4 .3 .2 .1		
	-	Eavors Empagliflozin		

Figure 4. Full trial cohort: effects of empagliflozin on weight and systolic BP overall and by key bioimpedance substudy prespecified subgroups. Study-average differences are adjusted for baseline values of the dependent variable (in continuous form) and for any differences in key baseline characteristics (categories of age, sex, diabetes, eGFR, urinary albumin-creatinine ratio, and region) between treatment groups and weighted in proportion to the amount of time between follow-up visits (see Supplemental Methods). Each analysis includes all individuals with at least one follow-up measurement of the outcome variable with mean imputation of missing baseline measurements. For comparison, between-group differences in the substudy cohort were -0.7 (95% Cl -1.3 to -0.1) kg and -3.3 (-5.5 to -1.2) mm Hg for weight and systolic BP, respectively.

mechanistic trials have reported plasma volume reductions by SGLT2 inhibitors²⁹ and raised a hypothesis that SGLT2 inhibitors reduce interstitial volume more than plasma volume.²⁸ Previously collected bioimpedance data in patients taking SGLT2 inhibitors are limited to mainly nonrandomized studies.^{30–33} To the best of our knowledge, the 16-week DECREASE trial provides the only peer reviewed published randomized evidence on the effects of SGLT2 inhibitors on bioimpedance parameters to date. It found that, in 66 participants with type 2 diabetes—CKD status not reported—dapagliflozin reduced extracellular fluid by approximately 1 L and systolic BP by approximately 4 mm Hg at 10 days versus placebo.³⁴ EMPA-KIDNEY now substantially extends these previous findings by studying longer term effects (over 18 months) in a much larger number of participants in a placebo-controlled trial.

Before the results of this substudy, attenuation of diuretic effects at low levels of kidney function was considered plausible as SGLT2 inhibitors have little effect on glycemia at lower eGFR due to attenuated levels of glycosuria.^{11,14,35–37} Despite

this, we found consistent effects on "Fluid Overload" across the eGFR-based subgroups. Similarly, effects did not vary by baseline fluid status, diuretic use, or albuminuria. These findings are analogous to results from large randomized trials in heart failure populations that included a large proportion of patients with CKD and low eGFR and demonstrated consistent effects of SGLT2 inhibitors on cardiovascular death or hospitalization for heart failure irrespective of sex, diabetes, eGFR, or NT-proBNP at baseline.¹³

It is also relevant that the effect of empagliflozin on fluid loss in EMPA-KIDNEY was achieved safely. Although estimates of ECW reduction reflected loss of ECW that is not considered to be in excess by the three-compartment model, there was no increased risk of participant reports of symptomatic dehydration in the full trial or substudy cohorts (Supplemental Table 12) nor any increased risk of AKI.¹¹

We also report assessments of the effects of empagliflozin on anthropometry, BP, HbA1c, and hematocrit for the full trial and substudy cohorts, with the full trial data providing better statistical power to assess for any effect modification between subgroups of participant. The effects of empagliflozin on weight and HbA1c in EMPA-KIDNEY are generally consistent with results from other CKD trials. Evaluation of the Effects of Canagliflozin on Renal and Cardiovascular Outcomes in Participants With Diabetic Nephropathy (CREDENCE) studied 4401 participants with type 2 diabetes and a mean eGFR of 56 ml/min per 1.73 m². Compared with placebo, the mean weight was 0.80 kg (95% CI, 0.69 to 0.92) lower in the canagliflozin group, and there was a relatively modest difference in HbA1c (-0.25%, 95% CI, -0.31 to -0.20).38 The Study to Evaluate the Effect of Dapagliflozin on Renal Outcomes and Cardiovascular Mortality in Patients With Chronic Kidney Disease trial studied 4304 participants with a mean eGFR of 43 ml/min per 1.73 m² and included 2996 participants with diabetes.³⁹ The between-group difference in HbA1c in those with diabetes was -1.1 mmol/mol (95% CI, -2.1 to 0.0).⁴⁰ The overall between-group difference in systolic BP in EMPA-KIDNEY of -2.6 mm Hg (95% CI, -3.3 to -1.9) was also similar to the other large CKD trials: CREDENCE difference -3.3 mm Hg (95% CI, -3.9 to -2.7)³⁸ and DAPA-CKD difference -2.9 mm Hg (95% CI, -3.6 to -2.3).41,42 In EMPA-KIDNEY, there were somewhat larger antihypertensive effects in participants with diabetes (heterogeneity P = 0.001). This pattern was not observed in bioimpedance-derived "Fluid Overload" analyses, raising the hypothesis that SGLT2 inhibition may have additional antihypertensive effects that are more prominent in patients with diabetes, and which are distinct from their diuretic effects (possibly through effects on vascular stiffness or endothelial function).43-45 The lack of measured effect of empagliflozin on adiposity is consistent with its modest effects on glycated hemoglobin observed in CKD populations.

Study Limitations

EMPA-KIDNEY demonstrated the clear benefits of SGLT2 inhibition on kidney disease progression in a wide range of

patients with CKD at risk of progression, including about a one-third reduction in the risk of needing to start kidney replacement therapy.¹¹ This large EMPA-KIDNEY substudy benefits from its sample size, long duration, systematic measurements, and randomized double-blind design. These help ensure between-group differences are unbiased and reliable. The BCM device has some technical limitations. For example, BCM parameters are derived, not direct measurements and are based on formulae normalized to healthy reference populations. Estimations may also be less accurate at extremes of "Fluid Overload" (although extremes of levels were uncommon in the substudy population). Furthermore, imprecision in fat mass estimates mean the lack of statistical effect on fat mass does not exclude some effect. The BCM device also does not reliably assess subtypes of adiposity (e.g., visceral versus peripheral). Follow-up was affected by coronavirus disease 2019 restrictions resulting in some missed bioimpedance measurements, and the prespecified key secondary composite analysis was underpowered because of lower cardiovascular risk in the trial population than was predicted during its design. Nevertheless, this substudy collected sufficient data to provide reliable and clear results for the primary and other continuously measured outcomes. Owing to the regions contributing to the substudy, Asian, Black, Mixed, and Other races were underrepresented, but effects on weight, HbA1c, and BP for the full trial cohort were broadly similar to the substudy results across the studied races, suggesting our conclusions are likely to be generalizable. Finally, the use of other diuretics was determined by local doctors and not controlled by the protocol. We observed more new use of loop diuretics among those allocated to placebo, so the presented estimates of effects on fluid parameters, weight, and BP may be slight underestimates of the full effect of empagliflozin.

In summary, the EMPA-KIDNEY bioimpedance substudy found that fluid excess is common in a broad population of patients with CKD at risk of progression and that empagliflozin resulted in sustained reductions in "Fluid Overload," weight, and BP in patients with CKD with and without diabetes, even in patients with low levels of kidney function.

DISCLOSURES

The EMPA-KIDNEY bioimpedance substudy was initiated, designed, conducted, analyzed, and reported by the University of Oxford with a steering committee of experts. This paper has not been published previously in whole or part. The Clinical Trial Service Unit and Epidemiological Studies Unit (Oxford, UK) have a staff policy of not accepting honorarium or other payments from the pharmaceutical industry, except for the reimbursement of costs to participate in scientific meetings (see https://www.ctsu.ox.ac.uk/ about/ctsu_honoraria_25june14–1.pdf). C. Baigent reports advisory or leadership role: Chair of the European Society of Cardiology Clinical Practice Guidelines Committee 2020–22, and other interests or relationships: trustee of the UK charity Alport-UK, which supports patients and families with Alport syndrome. S. Brenner reports participation on a nephrologist advisory board of Boehringer Ingelheim. V. Cejka reports support to attend meetings from Boehringer Ingelheim, Lilly, and the University of Oxford. V. Cejka also reports research funding: EMPA-KIDNEY study, PI in Wurzburg C. Wanner: The RENAL LIFECYCLE study(NCT05374291): PI in Wurzburg C. Wanner; HELIOS-B study (NCT04153149), PI in Wurzburg C. Morbach; CARDIO-TTRansform study (NCT04136171), and PI in Wurzburg S. Stork; honoraria: Pfizer; and other interests or relationships: financial support for travel and attendance of scientific congresses from Alnylam. D.Z.I. Cherney reports grant funding from AstraZeneca, Boehringer Ingelheim, CSL-Behring, Eli Lilly, Janssen, Merck & Co. Inc., Novo Nordisk, and Sanofi; and consulting fees from AbbVie, AstraZeneca, Bayer, BMS, Boehringer Ingelheim, CSL-Behring, Eli Lilly, Gilead, Inversago, Janssen, Lexicon, Maze, Merck & Co. Inc., Mitsubishi-Tanabe, Novartis, Novo Nordisk, Otsuka, Prometic, Sanofi, and Youngene. D.Z.I. Cherney also reports honoraria: AbbVie, AstraZeneca, Bayer, BMS, Boehringer Ingelheim-Lilly, CSL-Behring, Janssen, Maze, Merck, Mitsubishi-Tanabe, Novartis, Novo Nordisk, Otsuka, Prometic, Sanofi, and Youngene; and advisory or leadership role: AstraZeneca, Bayer, BMS, Boehringer Ingelheim-Lilly, CSL-Behring, Janssen, Lexicon, Maze, Merck, Novartis, and Novo Nordisk. S.J. Davies reports consultancy: Ellen Medical; research funding: Baxter Healthcare; honoraria: Baxter Healthcare and Fresenius Medical Care; and advisory or leadership role: International Society of Peritoneal Dialysis (Member, co-chair PDOPPS Committee), International Society of Nephrology (Kidney Failure Strategy), President EuroPD, and Trustee Kidney Research UK. J.B. Green reports grant funding from Boehringer Ingelheim, Lilly, Merck, and Roche; consulting fees from Anji, AstraZeneca, Bayer, Boehringer Ingelheim, Lilly, Merck, Novo Nordisk, Pfizer, Valo, and Vertex; honoraria for lectures from Boehringer Ingelheim; support to attend meetings from Bayer, Lilly, and Novo Nordisk; and writing support from Bayer. J.B. Green also reports research funding: Bluedrop; and honoraria: Anji, AstraZeneca, Bayer, Lilly, Novo Nordisk, Pfizer, Valo, and Vertex. W.G. Herrington was additionally funded by an MRC Kidney Research UK Professor David Kerr Clinician Scientist Award (MR/R007764/1). S.J. Hauske, D. Steubl, and M. Brückmann are employees of Boehringer Ingelheim International GmbH. W.G. Herrington also reports advisory or leadership role: NDT subject editor; UK Kidney Association, European Society of Cardiology & KDIGO guideline committee roles, and UK Renal Trial Network Chair. S.J. Hauske, D. Steubl, and M. Brückmann are employees of Boehringer Ingelheim International GmbH. L.S. Hooi reports support to attend meetings from the Malaysian Society of Nephrology and unpaid leadership roles as editor, the Malaysian Dialysis Transplant Registry; and chairman, Nephrology Medical Education Committee of the Malaysian Medical Council. L.S. Hooi reports other interests or relationships: editor, Malaysia Dialysis and Transplant Registry and Malaysian Society of Nephrology. P.K. Judge reports research funding: The UK HARP-III trial was funded by a grant to the University of Oxford from Novartis. D.F. Keane reports institutional grant funding from Baxter Healthcare for an Investigator Initiated Research Grant Award. D.F. Keane reports research funding: Baxter Healthcare Investigator Initiated Grant. M.J. Landray reports research funding: Apollo Tx, Boehringer Ingelheim, Moderna, Novartis, Regeneron, and Sanofi. J.S. Lees reports personal lectureship honoraria from AstraZeneca. P.B. Mark reports grant funding (paid to institution) from AstraZeneca and Boehringer Ingelheim; consulting fees (paid to institution) from Astellas, AstraZeneca, Bayer, Boehringer Ingelheim, and GSK; honoraria (paid to institution) from AstraZeneca, Boehringer Ingelheim, and Pharmacosmos; and personal payment from Novartis for participation on a Data Safety Monitoring Board or Advisory Board. P.B. Mark also reports consultancy: Pharmacosmos; and honoraria: Astellas, Bayer, GSK, and Pharmacosmos. K.J. Mayne reports research funding: MRC-UK-core funding paid to department. K.J. Mayne, N. Staplin, P.K. Judge, D. Preiss, J. Emberson, R. Dayanandan, R. Lee, J. Nolan, A. Omata, M.J. Landray, C. Baigent, R. Haynes, and W.G. Herrington report grant funding paid to their institution (the University of Oxford) from Boehringer Ingelheim and Eli Lilly, and funding from the United Kingdom Medical Research Council (MRC) (to the Clinical Trial Service Unit and Epidemiological Studies Unit; reference no., MC_UU_00017/3), the British Heart Foundation, National Institute for

Health and Care Research Biomedical Research Council, and Health Data Research (UK). R. Pontremoli reports honoraria from Alfa-Sigma, AstraZeneca, Boehringer Ingelheim, Lilly, Menarini, MSD, Novartis, and Novo Nordisk. D. Preiss reports research funding: Novartis and Novo Nordisk. N. Staplin reports research funding: Novo Nordisk; and advisory or leadership role: associate editor for Nephrology Dialysis Transplant. J. Stegbauer reports grant funding and honoraria from AstraZeneca and Boehringer Ingelheim; and support to attend meetings from Boehringer Ingelheim (Annual meeting of the German Society of Nephrology). J. Stegbauer also reports research funding: German Research Foundation; honoraria: Bayer Life Science; advisory or leadership role: editorial board: Hypertension and Kidney360; and other interests or relationships: German Society of Nephrology, AHA High Blood Pressure, and German Society of Hypertension. K.R. Tuttle reports grant funding from the National Institutes of Health, NIH (NIDDK, NHLBI, NCATS, and NIMHD), the Centers for Disease Control and Prevention (CDC), Bayer, and Travere; consulting fees from AstraZeneca, Bayer, Boehringer Ingelheim, Lilly, Novo Nordisk, Pfizer, and Travere; honoraria from AstraZeneca, Bayer, Lilly, and Novo Nordisk; support to attend meetings from Novo Nordisk; unpaid roles on Data Safety Monitoring/Advisory Boards for NIDDK and George Clinical; and unpaid leadership roles as Chair, Diabetic Kidney Disease Collaborative, American Society of Nephrology and Board of Directors, and Kidney Health Initiative. K.R. Tuttle reports consultancy: AstraZeneca, Bayer, Boehringer Ingelheim, Novo Nordisk, and Travere; research funding: Bayer and Travere; and honoraria: Bayer, Boehringer Ingelheim, and Novo Nordisk. C. Wanner reports institutional grant funding from Boehringer Ingelheim and Sanofi; consulting fees from AstraZeneca and Astellas, Bayer, and Boehringer Ingelheim; honoraria for lectures from Amgen, AstraZeneca, Bayer, Boehringer Ingelheim, CSL Vifor, Fresenius Medical Care, MSD, and Sanofi. C. Wanner also reports consultancy: CSL Vifor, GSK, MSD, and Novo Nordisk; honoraria: Amicus, Astellas, Chiesi, Eli Lilly, FMC, GSK, Novartis, Stadapharm, and Takeda; and other interests or relationships: European Renal Association (ERA).

FUNDING

The EMPA-KIDNEY bioimpedance substudy was funded by a grant to the University of Oxford from Boehringer Ingelheim (the sponsor) with additional funding from Eli Lilly. EMPA-KIDNEY was also supported by core funding to the Medical Research Council (MRC) Population Health Research Unit at the University of Oxford, which is part of the Clinical Trial Service Unit and Epidemiological Studies Unit (CTSU) (MRC grant numbers MC_UU_00017/3 and MC_UU_00017/ 4). CTSU also receives funding from the British Heart Foundation and Health Data Research (UK). D.F. Keane is supported by a Research Center grant from Science Foundation Ireland [grant number RC/ 2073_P2 (CÚRAM)]. W.G. Herrington was supported by a Medical Research Council–Kidney Research UK Professor David Kerr Clinician Scientist Award (MR/R007764/1).

ACKNOWLEDGMENTS

The authors thank the participants, the local site staff, regional coordinating center staff, and all members of EMPA-KIDNEY committees. The authors particularly thank the substudy participants and regional coordinating center teams in Germany (led by Dr Marcela Fajardo-Moser) and the United Kingdom (led by Yanru Qiao, supported by Mo Gray) for generating high-quality data during the COVID-19 pandemic. All authors are members of The EMPA-KIDNEY Bioimpedance Substudy Group.

AUTHOR CONTRIBUTIONS

Conceptualization: William G. Herrington, Johannes Stegbauer, Christoph Wanner.

Data curation: William G. Herrington, David F. Keane, Kaitlin J. Mayne, Natalie Staplin, Daniele Trinca.

Formal analysis: William G. Herrington, David F. Keane, Kaitlin J. Mayne, Natalie Staplin, Daniele Trinca.

Funding acquisition: Colin Baigent, Richard Haynes, William G. Herrington, Martin J. Landray.

Investigation: William G. Herrington, David F. Keane, Kaitlin J. Mayne, Natalie Staplin.

Methodology: William G. Herrington, David F. Keane, Kaitlin J. Mayne, Natalie Staplin.

Project administration: Rejive Dayanandan, William G. Herrington, Kaitlin J. Mayne, Ryonfa Lee, John Nolan, and Akiko Omata.

Resources: William G. Herrington, David F. Keane, Kaitlin J. Mayne, Daniele Trinca.

Software: David F. Keane, Kaitlin J. Mayne, John Nolan, Natalie Staplin, Daniele Trinca.

Supervision: William G. Herrington, Jennifer S. Lees, Patrick B. Mark, Natalie Staplin.

Validation: David F. Keane, Kaitlin J. Mayne, Natalie Staplin, Daniele Trinca.

Visualization: William G. Herrington, Kaitlin J. Mayne, Natalie Staplin.

Writing – original draft: William G. Herrington, Kaitlin J. Mayne, Natalie Staplin.

Writing – review & editing: Colin Baigent, Susanne Brenner, Martina Brückmann, Vladimir Cejka, David Z.I. Cherney, Simon J. Davies, Rejive Dayanandan, Jonathan Emberson, Jennifer B. Green, Sibylle J. Hauske, Richard Haynes, William G. Herrington, Lai Seong Hooi, Parminder K. Judge, David F. Keane, Martin J. Landray, Ryonfa Lee, Jennifer S. Lees, Patrick B. Mark, Kaitlin J. Mayne, John Nolan, Akiko Omata, Roberto Pontremoli, David Preiss, Natalie Staplin, Johannes Stegbauer, Dominik Steubl, Daniele Trinca, Katherine R. Tuttle, Christoph Wanner.

DATA SHARING STATEMENT

The complete deidentified patient dataset used for presented analyses will be available in due course and the application system to apply to use data will open 6 months after publication. Departmental policy details can be found here https://www.ndph.ox.ac.uk/data-access. In adherence with the Boehringer Ingelheim Policy on Transparency and Publication of Clinical Study Data, scientific and medical researchers can request access to clinical study data after publication of the primary manuscript and secondary analyses in peer-reviewed journals and regulatory and reimbursement activities are completed, normally within 1 year after the marketing application has been granted by major regulatory authorities. Researchers should use the https://vivli.org/link to request access to study data and visit https://www.mystudywindow.com/msw/datasharing for further information.

SUPPLEMENTAL MATERIAL

This article contains the following supplemental material online at http://links.lww.com/JSN/E563.

Members of the EMPA-KIDNEY Collaborative Group.

Supplemental Methods.

Substudy Protocol Supplement.

Substudy Data Analysis Plan.

Supplemental Table 1. Bioimpedance substudy cohort: additional baseline characteristics.

Supplemental Table 2. Bioimpedance substudy cohort: baseline characteristics by categories of baseline bioimpedance-derived "Fluid Overload."

Supplemental Table 3. Baseline characteristics for the substudy, substudy region, and the full trial cohorts.

Supplemental Table 4. Sensitivity analyses for the effects of empagliflozin on mean bioimpedance-derived absolute "Fluid Overload" in L.

Supplemental Table 5. Bioimpedance substudy cohort: unadjusted baseline means and adjusted study averages for "Fluid Overload" for each subgroup by treatment group (additional data to accompany Figure 3).

Supplemental Table 6. Effects of empagliflozin on tertiary (and *post hoc*) bioimpedance-derived parameters assessments by time.

Supplemental Table 7. Effects of empagliflozin on other bioimpedancederived adiposity parameters.

Supplemental Table 8. Effects of empagliflozin on weight and body mass index (bioimpedance substudy and full trial cohorts).

Supplemental Table 9. Effects of empagliflozin on waist and hip measurements (bioimpedance substudy and full trial cohorts).

Supplemental Table 10. Effects of empagliflozin on glycated hemoglobin and hematocrit (bioimpedance substudy and full trial cohorts).

Supplemental Table 11. Effects of empagliflozin on BP (bioimpedance substudy and full trial cohorts).

Supplemental Table 12. Bioimpedance substudy cohort: effects of empagliflozin on dehydration by categories of baseline bioimpedance-derived "Fluid Overload."

Supplemental Figure 1. Bioimpedance substudy cohort CONSORT flow chart.

Supplemental Figure 2. Effects of empagliflozin on mean absolute "Fluid Overload" (bioimpedance substudy cohort: *post hoc* subgroups).

Supplemental Figure 3. Correlation between change in weight (relative to baseline) with change in different bioimpedance indices at the 2 month follow-up visit, by treatment allocation.

Supplemental Figure 4. Effects of empagliflozin on weight, body mass index, systolic BP, glycated hemoglobin, and hematocrit by race (full trial cohort).

REFERENCES

- Tonelli M, Muntner P, Lloyd A, et al. Risk of coronary events in people with chronic kidney disease compared with those with diabetes: a population-level cohort study. *Lancet.* 2012;380(9844):807–814. doi: 10.1016/S0140-6736(12)60572-8
- Go AS, Chertow GM, Fan D, McCulloch CE, Hsu C-Y. Chronic kidney disease and the risks of death, cardiovascular events, and hospitalization N Engl J Med. 2004;351(13):1296–1305. doi:10.1056/ NEJMoa041031
- Park M, Hsu CY, Li Y, et al. Associations between kidney function and subclinical cardiac abnormalities in CKD. J Am Soc Nephrol. 2012; 23(10):1725–1734. doi:10.1681/ASN.2012020145
- Foley RN, Parfrey PS, Kent GM, Harnett JD, Murray DC, Barre PE. Longterm evolution of cardiomyopathy in dialysis patients. *Kidney Int.* 1998; 54(5):1720–1725. doi:10.1046/j.1523-1755.1998.00154.x
- Suzuki T, Agarwal SK, Deo R, et al. Kidney function and sudden cardiac death in the community: the Atherosclerosis Risk in Communities (ARIC) Study. Am Heart J. 2016;180:46–53. doi:10.1016/j.ahj.2016.07. 004
- Chronic Kidney Disease Prognosis Consortium, Matsushita K, van der Velde M, Astor BC, et al. Association of estimated glomerular filtration rate and albuminuria with all-cause and cardiovascular mortality in general population cohorts: a collaborative meta-analysis. *Lancet.* 2010;375(9731):2073–2081. doi:10.1016/S0140-6736(10)60674-5

- Pecoits-Filho R, Gonçalves S, Barberato SH, et al. Impact of residual renal function on volume status in chronic renal failure. *Blood Purif.* 2004;22(3):285–292. doi:10.1159/000078699
- Mayne KJ, Shemilt R, Keane DF, Lees JS, Mark PB, Herrington WG. Bioimpedance indices of fluid overload and cardiorenal outcomes in heart failure and chronic kidney disease: a systematic review. J Card Fail. 2022;28(11):1628–1641. doi:10.1016/j.cardfail.2022.08.005
- Chamney PW, Wabel P, Moissl UM, et al. A whole-body model to distinguish excess fluid from the hydration of major body tissues. Am J Clin Nutr. 2007;85(1):80–89. doi:10.1093/ajcn/85.1.80
- Zoccali C, Moissl U, Chazot C, et al. Chronic fluid overload and mortality in ESRD. J Am Soc Nephrol. 2017;28(8):2491–2497. doi:10. 1681/ASN.2016121341
- The EMPA-KIDNEY Collaborative Group, Herrington WG, Staplin N, Green JB, et al. Empagliflozin in patients with chronic kidney disease. N Engl J Med. 2023;388(2):117–127. doi:10.1056/NEJMoa2204233
- Nuffield Department of Population Health Renal Studies Group.SGLT2 inhibitor Meta-Analysis Cardio-Renal Trialists' Consortium. Impact of diabetes on the effects of sodium glucose co-transporter-2 inhibitors on kidney outcomes: collaborative meta-analysis of large placebocontrolled trials. *Lancet.* 2022;400(10365):1788–1801. doi:10.1016/ S0140-6736(22)02074-8
- Vaduganathan M, Docherty KF, Claggett BL, et al. SGLT-2 inhibitors in patients with heart failure: a comprehensive meta-analysis of five randomised controlled trials. *Lancet.* 2022;400(10354):757–767. doi: 10.1016/S0140-6736(22)01429-5
- Macha S, Mattheus M, Halabi A, Pinnetti S, Woerle HJ, Broedl UC. Pharmacokinetics, pharmacodynamics and safety of empagliflozin, a sodium glucose cotransporter 2 (SGLT2) inhibitor, in subjects with renal impairment. *Diabetes Obes Metab.* 2014;16(3):215–222. doi:10.1111/ dom.12182
- Heerspink HJ, Perkins BA, Fitchett DH, Husain M, Cherney DZ. Sodium glucose cotransporter 2 inhibitors in the treatment of diabetes mellitus: cardiovascular and kidney effects, potential mechanisms, and clinical applications. *Circulation*. 2016;134(10):752–772. doi:10.1161/ CIRCULATIONAHA.116.021887
- EMPA-KIDNEY Collaborative Group. Design, recruitment, and baseline characteristics of the EMPA-KIDNEY trial. Nephrol Dial Transplant. 2022;37(7):1317–1329. doi:10.1093/ndt/gfac040
- Tabinor M, Elphick E, Dudson M, Kwok CS, Lambie M, Davies SJ. Bioimpedance-defined overhydration predicts survival in end stage kidney failure (ESKF): systematic review and subgroup meta-analysis. *Sci Rep.* 2018;8(1):4441. doi:10.1038/s41598-018-21226-y
- Wabel P, Chamney P, Moissl U, Jirka T. Importance of whole-body bioimpedance spectroscopy for the management of fluid balance. *Blood Purif.* 2009;27(1):75–80. doi:10.1159/000167013
- Wabel P, Chamney P, Moissl U, et al. Reproducibility of bioimpedance spectroscopy (BIS) for the assessment of body composition and dry weight. J Am Soc Nephrol. 2007:255.
- Davies SJ, Coyle D, Lindley EJ, et al. Bio-impedance spectroscopy added to a fluid management protocol does not improve preservation of residual kidney function in incident hemodialysis patients in a randomized controlled trial. *Kidney Int.* 2023;104(3):587–598. doi:10. 1016/j.kint.2023.05.016
- Moissl UM, Wabel P, Chamney PW, et al. Body fluid volume determination via body composition spectroscopy in health and disease. *Physiol Meas.* 2006;27(9):921–933. doi:10.1088/0967-3334/27/9/012
- Wabel P, Moissl U, Chamney P, et al. Towards improved cardiovascular management: the necessity of combining blood pressure and fluid overload. Nephrol Dial Transplant. 2008;23(9):2965–2971. doi:10. 1093/ndt/gfn228
- Van Biesen W, Williams JD, Covic AC, et al. Fluid status in peritoneal dialysis patients: the European Body Composition Monitoring (EuroBCM) study cohort. *PLoS One*. 2011;6(2):e17148. doi:10.1371/ journal.pone.0017148

- Siriopol D, Siriopol M, Stuard S, et al. An analysis of the impact of fluid overload and fluid depletion for all-cause and cardiovascular mortality. *Nephrol Dial Transplant*. 2019;34(8):1385–1393. doi:10.1093/ndt/ gfy396
- Ward LC, Essex T, Cornish BH. Determination of Cole parameters in multiple frequency bioelectrical impedance analysis using only the measurement of impedances. *Physiol Meas.* 2006;27(9):839–850. doi: 10.1088/0967-3334/27/9/007
- Herrington WG, Savarese G, Haynes R, et al. Cardiac, renal, and metabolic effects of sodium-glucose co-transporter 2 inhibitors: a position paper from the European Society of Cardiology ad-hoc task force on sodium-glucose co-transporter 2 inhibitors. *Eur J Heart Fail*. 2021;23(8):1260–1275. doi:10.1002/ejhf.2286
- Inzucchi SE, Zinman B, Fitchett D, et al. How does empagliflozin reduce cardiovascular mortality? Insights from a mediation analysis of the EMPA-REG OUTCOME trial. *Diabetes Care*. 2018;41(2):356–363. doi: 10.2337/dc17-1096
- Hallow KM, Helmlinger G, Greasley PJ, McMurray JJV, Boulton DW. Why do SGLT2 inhibitors reduce heart failure hospitalization? A differential volume regulation hypothesis. *Diabetes Obes Metab.* 2018; 20(3):479–487. doi:10.1111/dom.13126
- Lambers Heerspink HJ, de Zeeuw D, Wie L, Leslie B, List J. Dapagliflozin a glucose-regulating drug with diuretic properties in subjects with type 2 diabetes. *Diabetes Obes Metab.* 2013;15(9): 853–862. doi:10.1111/dom.12127
- Sen T, Scholtes R, Greasley PJ, et al. Effects of dapagliflozin on volume status and systemic haemodynamics in patients with chronic kidney disease without diabetes: results from DAPASALT and DIAMOND. *Diabetes Obes Metab.* 2022;24(8):1578–1587. doi:10.1111/dom. 14729
- Scholtes RA, Muskiet MHA, van Baar MJB, et al. Natriuretic effect of two weeks of dapagliflozin treatment in patients with type 2 diabetes and preserved kidney function during standardized sodium intake: results of the DAPASALT trial. *Diabetes Care*. 2021;44(2):440–447. doi: 10.2337/dc20-2604
- Schork A, Eberbach M-L, Bohnert BN, et al. SGLT2 inhibitors decrease overhydration and proteasuria in patients with chronic kidney disease: a longitudinal observational study. *Research Square*. 2023. doi:10. 21203/rs.3.rs-3343672/v1
- Schork A, Bohnert BN, Heyne N, Birkenfeld AL, Artunc F. Overhydration measured by bioimpedance spectroscopy and urinary serine protease activity are risk factors for progression of chronic kidney disease. *Kidney Blood Press Res.* 2020;45(6):955–968. doi:10.1159/ 000510649
- 34. van Ruiten CC, Smits MM, Kok MD, et al. Mechanisms underlying the blood pressure lowering effects of dapagliflozin, exenatide, and their combination in people with type 2 diabetes: a secondary analysis of a randomized trial. *Cardiovasc Diabetol.* 2022;21(1):63. doi:10.1186/ s12933-022-01492-x
- Petrykiv S, Sjöström CD, Greasley PJ, Xu J, Persson F, Heerspink HJL. Differential effects of dapagliflozin on cardiovascular risk factors at varying degrees of renal function. *Clin J Am Soc Nephrol.* 2017;12(5): 751–759. doi:10.2215/CJN.10180916
- Kohan DE, Fioretto P, Tang W, List JF. Long-term study of patients with type 2 diabetes and moderate renal impairment shows that dapagliflozin reduces weight and blood pressure but does not improve glycemic control. *Kidney Int.* 2014;85(4):962–971. doi:10. 1038/ki.2013.356
- Cherney DZI, Cooper ME, Tikkanen I, et al. Pooled analysis of phase III trials indicate contrasting influences of renal function on blood pressure, body weight, and HbA1c reductions with empagliflozin. *Kidney Int.* 2018;93(1):231–244. doi:10.1016/j.kint.2017.06.017
- Perkovic V, Jardine MJ, Neal B, et al. Canagliflozin and renal outcomes in type 2 diabetes and nephropathy. N Engl J Med. 2019;380(24): 2295–2306. doi:10.1056/NEJMoa1811744

- Heerspink HJL, Stefánsson BV, Correa-Rotter R, et al. Dapagliflozin in patients with chronic kidney disease. N Engl J Med. 2020;383(15): 1436–1446. doi:10.1056/NEJMoa2024816
- Persson F, Rossing P, Vart P, et al. Efficacy and safety of dapagliflozin by baseline glycemic status: a prespecified analysis from the DAPA-CKD trial. *Diabetes Care*. 2021;44(8):1894–1897. doi:10. 2337/dc21-0300
- Provenzano M, Toto RD, Vart P, et al. POS-255 effect of dapagliflozin on blood pressure in patients with CKD: a pre-specified analysis from DAPA-CKD. *Kidney Int Rep.* 2022;7(2):S112. doi:10.1016/j.ekir.2022. 01.274
- 42. Ye N, Jardine MJ, Oshima M, et al. Blood pressure effects of canagliflozin and clinical outcomes in type 2 diabetes and chronic kidney

disease: insights from the CREDENCE trial. *Circulation*. 2021;143(18): 1735–1749. doi:10.1161/CIRCULATIONAHA.120.048740

- Lytvyn Y, Kimura K, Peter N, et al. Renal and vascular effects of combined SGLT2 and angiotensin-converting enzyme inhibition. *Circulation*. 2022; 146(6):450–462. doi:10.1161/CIRCULATIONAHA.122.059150
- Lytvyn Y, Bjornstad P, Udell JA, Lovshin JA, Cherney DZI. Sodium glucose cotransporter-2 inhibition in heart failure: potential mechanisms, clinical applications, and summary of clinical trials. *Circulation*. 2017; 136(17):1643–1658. doi:10.1161/CIRCULATIONAHA.117.030012
- 45. Cherney DZ, Perkins BA, Soleymanlou N, et al. The effect of empagliflozin on arterial stiffness and heart rate variability in subjects with uncomplicated type 1 diabetes mellitus. *Cardiovasc Diabetol.* 2014; 13:28. doi:10.1186/1475-2840-13-28

AFFILIATIONS

¹Medical Research Council Population Health Research Unit, Clinical Trial Service Unit and Epidemiological Studies Unit (CTSU), Nuffield Department of Population Health, University of Oxford, Oxford, United Kingdom

²School of Cardiovascular and Metabolic Health, College of Medical and Veterinary Life Sciences, University of Glasgow, Glasgow, United Kingdom

³CÚRAM SFI Research Centre for Medical Devices, HRB-Clinical Research Facility Galway, National University of Ireland Galway, Galway, Ireland ⁴Würzburg University Clinic, Würzburg, Germany

⁵Department of Nephrology, Medical Faculty, University Hospital Düsseldorf, Heinrich-Heine-University Düsseldorf, Düsseldorf, Germany

⁶CARID, Cardiovascular Research Institute Düsseldorf, Medical Faculty and University Hospital Düsseldorf, Heinrich-Heine-University Düsseldorf, Dusseldorf, Germany

⁷Oxford Kidney Unit, Oxford University Hospitals NHS Foundation Trust, Oxford, United Kingdom

⁸Leeds Teaching Hospitals NHS Trust, Leeds, United Kingdom

⁹Duke Clinical Research Institute, Durham, North Carolina

¹⁰University of Toronto, Toronto, Canada

¹¹Department of Medicine and Haemodialysis Unit, Sultanah Aminah Hospital, Johor Bahru, Malaysia

¹²Università degli Studi and IRCCS Ospedale Policlinico San Martino di Genova, Genoa, Italy

¹³Providence Inland Northwest Health, University of Washington, Spokane, Washington

¹⁴School of Medicine, Keele University, Newcastle, United Kingdom

¹⁵Boehringer Ingelheim International GmbH, Ingelheim upon Rhein, Germany

¹⁶The Fifth Department of Medicine, University Medical Center Mannheim, Mannheim, Germany

¹⁷University of Heidelberg, Mannheim, Germany

¹⁸Department of Nephrology, Hospital Rechts der Isar, Technical University of Munich, Munich, Germany

¹⁹The First Department of Medicine, University Medical Center Mannheim, Mannheim, Germany